

This is a repository copy of Mapping SME productivity research: a systematic review of empirical evidence and future research agenda.

White Rose Research Online URL for this paper: https://eprints.whiterose.ac.uk/169756/

Version: Accepted Version

Article:

Owalla, B. orcid.org/0000-0001-9706-3144, Gherhes, C., Vorley, T. et al. (1 more author) (2022) Mapping SME productivity research: a systematic review of empirical evidence and future research agenda. Small Business Economics, 58 (3). pp. 1285-1307. ISSN 0921-898X

https://doi.org/10.1007/s11187-021-00450-3

This is a post-peer-review, pre-copyedit version of an article published in Small Business Economics. The final authenticated version is available online at: http://dx.doi.org/10.1007/s11187-021-00450-3.

Reuse

Items deposited in White Rose Research Online are protected by copyright, with all rights reserved unless indicated otherwise. They may be downloaded and/or printed for private study, or other acts as permitted by national copyright laws. The publisher or other rights holders may allow further reproduction and re-use of the full text version. This is indicated by the licence information on the White Rose Research Online record for the item.

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



Mapping SME Productivity Research: A Systematic Review of Empirical Evidence and

Future Research Agenda

Abstract

SMEs are the lifeblood of economies around the world. They play an important role in productivity growth, which

is crucial for developed economies as they adjust to major trends such as the industrial revolution, an aging

population, and changes in the nature of work. This study maps the SME productivity research landscape by way

of a systematic literature review focusing on the direct, indirect, as well as mediating/moderating factors that

enable or constrain productivity in SMEs. We review 109 empirical studies and highlight the fragmented nature

of the extant research in this field. Our thematic analysis identifies six key themes, namely organizational

environment, organizational capabilities, investments, types of innovation, external knowledgebase and

commercialization. By taking stock of existing knowledge, we highlight critical gaps and methodological issues

that limit our understanding of SME productivity. We propose a future research agenda to address current

shortcomings and advance knowledge on this topic. Implications for policy are also discussed.

Keywords: SMEs, Productivity, Systematic literature review, SME policy

JEL Classifications: D24 L50 O40

1

1 Introduction

Productivity, defined simply as the total output (e.g. goods, services) that can be produced with given inputs (e.g. labour, capital, resources), is a key driver of economic growth, social prosperity, and living standards. However, differing patterns and causes of productivity slowdown at the firm level have led to growing inequality in productivity since the 2008-2009 financial crisis (OECD 2019). For example, using firm level data for OECD countries, Criscuolo (2018) finds that the gap in productivity levels between firms in the top and bottom deciles of the productivity distribution increased by 14% between 2001 and 2012. This gap is partly attributed to an increase in divergence between technological frontier and the non-frontier firms (Andrews et al. 2016). Similar patterns are found in the productivity distribution of large and small firms with, for example, manufacturing SMEs achieving persistently lower productivity levels (only 62% in the case of small firms and 75% for medium firms) than large firms (OECD 2018). It is therefore essential to capture the heterogeneity in productivity growth amongst different firm size groups in order to better inform future research as well as policies targeted at reducing barriers to, and capitalizing on, opportunities for productivity growth (OECD 2017a).

In this systematic literature review, we focus on factors affecting productivity in small and medium sized enterprises (SMEs) as opposed to larger firms. The rationale for this is threefold. First, SMEs are the backbone of national economies worldwide, contributing significantly to economic growth and job creation, while at the same time having the potential to facilitate inclusive growth as economies adapt to major trends in the economy (Blancher et al. 2019; Motta 2020). For example, micro enterprises alone account for about 70 to 95% of all enterprises in OECD economies, and are a major source of employment and economic growth (OECD 2017a). Second, SMEs have been disproportionately affected by the 2008-2009 financial crisis, which also marks the widening gap in productivity growth between SMEs and large firms (OECD 2017b). In fact, studies indicate that despite the opportunities to participate in the global economy that digital technologies and global value chains provide, SMEs still lag behind in the digital transition, and are more adversely affected by structural barriers (Cusmano et al. 2018). Third, SMEs face unique constraints which make growth and productivity gains more challenging than in larger firms. In particular, small firms are constrained by the 'liability of smallness', a concept coined by Aldrich and Auster (1986) to explain the set of constraints stemming from: the lack of or uncertainty over financial resources; the impossibility to attract skilled workforce compared to large firms; the difficulty in meeting high interest rate payments and/or administrative costs generated by compliance with government regulations. Furthermore, small firms tend to be owner-manager centric, which restricts their ability to develop and execute strategies (Borch and Madsen 2007). For example, decision-making processes are constrained by the owner-manager's time, resources, information, and managerial skills (Gherhes et al. 2016; Osiyevskyy and Dewald 2015). Their tendency to focus on short-term survival also limits performance and detracts from demands of longer-term growth planning and strategizing (Beaver and Prince 2004). All these aspects have key implications for SME performance, and highlight the need to understand the specific challenges faced by SMEs in achieving productivity growth.

While interest in SME research boomed in the 1990s, research on SME productivity in particular gained increased attention from the mid-2000s when many western economies began experiencing a decline in labour productivity growth (Bughin et al. 2018; Gibb 2000). Nevertheless, while numerous studies have been carried out in entrepreneurship and management fields on different aspects of SME productivity, this knowledge remains highly fragmented. Previous systematic reviews on firm productivity have analysed the impact of Information Technology (Brynjolfsson and Yang 1996), work-family policy (Kossek and Ozeki 1999), manufacturing systems (Muthiah and Huang 2006), and structural change (Krüger 2008) on productivity. Those focused on SMEs have examined specific aspects related to productivity, such as the impact of performance management systems (Garengo et al. 2005), the relationship between innovation, exportation and growth (Love and Roper 2015), SME internationalization (Dabić et al. 2019; Paul et al. 2017), as well as the impact of cash holdings (La Rocca et al. 2019) and public investments (Dvouletý et al. 2020) on SME performance. While these reviews have advanced knowledge on the topic, they do not capture the broader landscape. As Cicea et al. (2019) note, the majority of empirical studies on SME performance - a key dimension of SME productivity - examine organisational factors in isolation, thereby overlooking the external environment in which SMEs operate and the interrelated factors which enable or constrain SME productivity. Moreover, factors influencing firm level productivity such as human capital, management practices, technology, fixed assets etc., are interdependent and cannot be understood in isolation (Prowle et al. 2017).

We therefore adopt a holistic approach to mapping SME productivity literature by carrying out an evidence-based systematic review of empirical studies that explicitly focus on SME productivity in western economies. The main aim of our study is to provide an integrated understanding of the main factors influencing SME productivity. In order to meaningfully explore the current landscape, we address the following research question: What are the direct, indirect, as well as mediating/moderating factors that enable or constrain productivity in SMEs? To the best of our knowledge, this is the first review that adopts such a holistic approach in attempting to map the fragmented SME productivity literature. Notably, the factors and mediators/moderators identified in this systematic review are neither specific to, nor restricted to SMEs, but rather provide a map of the current SME productivity research landscape that helps advance our understanding of the field by highlighting key gaps and areas for future research.

Through our systematic review of 109 empirical studies, we highlight the fragmented nature of SME productivity research landscape, and make the following contributions to the field. First, we thematically map empirical studies into six key themes organised around internal (*organizational environment, organizational capabilities, investments* and *types of innovation*) and external (*external knowledgebase, commercialization*) factors influencing SME productivity. Second, our descriptive and thematic analyses highlight critical gaps in existing knowledge, as well as methodological issues that currently restrict our understanding of how different factors such as technology, R&D, leadership skills, networks, and collaborations, interact to affect SME productivity. Third, we develop a research agenda and propose four main areas where future research could make useful contributions in advancing knowledge on SME productivity. These are the need for greater comparative analyses (i.e. within SMEs as well as between SMEs and large firms), the need to adopt a holistic approach to

understanding human capital in SMEs, the importance of context, and methodological developments that would better enable the analysis of underlying factors affecting SME productivity.

The remainder of the paper is organised as follows: section 2 explains the systematic review methodology and summarises the results of the database search; section 3 presents a descriptive analysis followed by a thematic analysis where we unpack and discuss the findings; section 4 discusses the results, sets a future research agenda, and considers the limitations of our study; section 5 discusses policy implications; and section 6 concludes.

2 Method

In this section, we summarize the systematic review methodology employed in our study. Systematic reviews offer a more transparent, scientific, and reproducible procedure for carrying out literature searches and analyses than narrative reviews (Hu et al. 2015; Tranfield et al. 2003). An evidence-based systematic review is therefore appropriate for fulfilling the aim of our study, enabling us to review empirical studies that explicitly focus on SME productivity in western developed economies. Our review consequently includes studies focusing on Europe (excluding transition economies), the UK, USA, Canada, Australia, and New Zealand. As such, empirical studies focusing on transition or developing economies (Abor et al. 2014; De Loecker 2007; Kim and Ro 2017; Nyikos et al. 2020; Ur Rehman et al. 2019), or those primarily focused on other aspects like SMEs efficiency, performance or competitiveness (Hamilton and Asundi 2010; Hogan and Coote 2014; Ipinnaiye et al. 2017; Siqueira and Cosh 2008), are considered outside the scope of this review. While the focus on western economies restricts our understanding of SME productivity to this specific context, the exclusion of other geographical regions was deemed necessary to ensure greater homogeneity, and allow us to take into consideration the effects of macrolevel factors on SME productivity (Ostapenko 2015; Rangamohan et al. 2007). The evidence-based approach also enabled us to provide an integrated understanding of factors influencing SME productivity that can better inform future research on the topic, as well as practitioners and policymakers' decisions with regard to improving productivity growth.

Similar to previous systematic reviews (Hu et al. 2015; Mallett et al. 2018; Thorpe et al. 2005), we follow Tranfield et al.'s (2003) rigorous systematic review methodology that includes three stages, namely planning, conducting and reporting. An overview of the systematic review process is presented in Figure 1 in Online Resource 1. In the planning stage, search strings were constructed in line with the aim of the review and based on a combination of key words related to SMEs and productivity (see Table 1 in Online Resource 1). In conducting the review, we identified and queried four databases, namely Scopus, ABI-Inform/ProQuest, Emerald and Science Direct. If a search generated less than 100 citations, the search string was amended to broaden the search scope. This initial search yielded a total of 1,530 citations. An initial analysis was then carried out to remove duplicates or articles that did not meet the inclusion criteria (see Table 2 in Online Resource 1). Further, in order to ensure scientific quality, only peer-reviewed articles published in academic journals were included (Jones et al. 2011; Liñán and Fayolle 2015). These analyses were carried out independently by two authors of this paper, resulting in a sample of 124 citations (81 articles were selected by both authors independently). Several discussions were

then held to decide which of the 43 articles that had been uniquely selected by either one of the authors should be included in the final review. To ensure that the review was comprehensive and did not exclude any potentially relevant articles, a review update was carried out in addition to the systematic search to identify articles published after our initial search or that the review might have missed. This comprehensive approach involved narrative searches on Google Scholar, as well as further searches at a later stage based on recommendations received from the anonymous reviewers during the review stage, and as a result of consultations with productivity experts. The narrative searches yielded an additional 24 articles. In the next step, the articles were classified into three categories based on their relevance i.e. articles that were directly relevant to the scope of the review (i.e. focused on factors that directly influence SME productivity), articles that were partially relevant (i.e. focused on factors that indirectly influence SME productivity), and articles that were less relevant (i.e. SME productivity is not the main focus, but is rather mentioned incidentally). Our final sample includes articles from the first two categories, resulting in a total of 109 relevant articles.

The last stage included carrying out a descriptive and thematic analysis of the review articles. In order to facilitate the descriptive analyses, we created a spreadsheet to capture information based on a reading guide adapted from Henry et al.'s (2016) systematic review (see Table 3 in Online Resource 1). An abductive coding process was used to carry out the thematic analyses, which involved the use of NVivo coding software. First order codes were inductively identified from the review articles. During the next stage, an iterative process, which involved moving back and forth between the data and theoretical literature, was followed in order to generate the second order analytical themes (Kennedy 2018; Miles et al. 2014). This iterative process involved consultations amongst the authors to further refine initial themes identified. The six main themes identified through this iterative process, were further categorized into aggregate dimensions of internal and external factors influencing SME productivity. An overview of the analytical themes and aggregate dimensions is presented in Table 4 in Online Resource 1.

2.1 Defining SMEs and productivity measures

While there is no single agreed upon definition of SMEs, these are generally considered to be non-subsidiary independent firms that employ fewer than a given number of employees, with the upper limit varying between 250 employees (e.g. in the European Union) and 500 employees (e.g. in the United States) respectively (OECD 2005). In this study, we use the OECD definition of SMEs, namely enterprises with fewer than 250 employees that comprise: micro (fewer than 10 employees), small (10–49 employees), and medium-sized (50–249 employees) enterprises (OECD 2017a).

Additionally, productivity is broadly understood in relation to the two most commonly used productivity measures, namely labour and total factor productivity. Labour productivity is a key determinant of living standards (i.e. per capita income) and was considered the main productivity measure until the 1950s when works by Solow (1957) resulted in the prevalent use of the Cobb-Douglas function and total (multi) factor productivity (TFP) (Godin 2009). TFP measures help to disentangle the direct growth contributions of labour, capital, intermediate

inputs, and technology; and is an important tool for reviewing growth patterns and assessing the potential for future economic growth. While the former measure is criticised for attributing all increases in efficiency to one factor (Linna et al. 2010), the latter presents a number of methodological issues that complicates the measurement of TFP (Van Beveren 2012). Despite these limitations, labour productivity and TFP are two of the most common productivity measures, and thus our systematic review includes empirical studies using either measure.

3 Results of the systematic review

This section presents the results of our systematic review in relation to the descriptive and thematic analyses carried out on the reviewed empirical studies. The descriptive analysis is presented first, followed by the thematic analysis.

3.1 Descriptive analysis

The 109 articles included in the review covered empirical studies published between 1994 to 2020 (inclusive), with the majority of studies (72%) being published within the last decade. The rising trend in number of articles highlights the traction gained by the concept of productivity, as well as the increased focus on SME productivity in both academic and policy discourses in recent years. Critically, we also find that the productivity measures, SME definitions, sample types and sizes, research methods, and data analysis techniques used in the reviewed studies, vary considerably. We summarize below the results of the descriptive analyses, with a more detailed description being provided in Table 5 in Online Resource 1.

The majority of reviewed studies (88%) focus on a single country, and only 4 of those focused on multiple countries specifically carry out cross-country comparisons in their analyses (see Figure 2 in Online Resource 1). Similarly, studies examine a variety of sectors ranging from manufacturing, construction, automotive, agriculture to information technology, textiles, food sector etc., with only 2 studies (Black and Lynch 1996; Yazdanfar 2013) carrying out sector-specific comparisons, and another 4 studies comparing both firm size and sectors (Antonelli et al. 2015; Falk and Hagsten 2015a; Lejarraga and Oberhofer 2015; Parisi et al. 2006). We also find that reviewed studies carrying out comparisons based on firm size and/or related factors such as age, absorptive capacity or region, are most common (22%), followed by those analysing the impact of different interventions (8%) and type/level of technology (6%).

Furthermore, the majority of reviewed studies are quantitative (86%), with sample sizes ranging from one case study (Esan et al. 2013) to 1,897,288 observations from a panel dataset (Foreman-Peck and Nicholls 2013, 2015). Data sources are diverse and include independent surveys (via mail, web, or telephone), data from the national statistical offices, balance sheet data, unstructured/semi-structured interviews, and case studies. The data analysis methods utilized are equally varied and include regression analysis, CDM models, structural equation modelling, case study analysis and observations. In addition, only 31% of reviewed studies address the issue of endogeneity. These studies use a variety of methods, such as the use of longitudinal data, lagged or predicted variables, the

Hausmann-Wu test, CDM framework, or the Generalised Method of Moments system estimator to control for it. Moreover, only 11% of the studies are longitudinal, and analyse the impact of factors such as process innovation, HRM practices, technology adoption, export participation or public subsidies on SME productivity.

Importantly, this variation in data sources and methods of analysis hinders the direct comparability of findings from reviewed studies. Nevertheless, while we acknowledge this as one of the limitations of the systematic review, our aim is to provide an integrated understanding of the main factors influencing SME productivity. We discuss these methodological challenges in section 4 where we also highlight areas for further consideration in future research.

3.2 Thematic analysis

Our thematic analyses identified six key themes which were further aggregated into two main groups, namely internal (*organizational environment; organizational capabilities; investments; type of innovation*) and external (*external knowledgebase; commercialization*) factors influencing SME productivity (see Table 4 in Online Resource 1). The categories and themes identified are not mutually exclusive, as majority of the reviewed studies (56%) are classified under more than one category/theme. Our analysis in this section therefore focuses on providing an integrated understanding of the different factors directly or indirectly influencing SME productivity. A mapping of the factors and related mediators/moderators directly influencing SME productivity, as well as a summary of key study findings, are presented in Figure 3 and Table 6 respectively in Online Resource 1.

Theme I - Organizational environment: this theme includes research focused on understanding how internal aspects such as workplace practices, ownership structure, industrial relations etc., affect a firm's operations. Reviewed studies analyse how the work environment and human resource (HR) policies impact a firm's labour productivity, with many studies indicating a positive effect of proactive HR planning and recruitment (Black and Lynch 1996; Koch and McGrath 1996; Lodefalk and Tang 2018; McGuinness et al. 2008). For example, Koch and McGrath (1996) find that proactively planning and recruiting the right people from the onset not only results in greater labour productivity, but also benefits the firm through the interaction effect between HR policies and capital intensity (i.e. ratio of assets to total employees). This effect exerts a more dominant influence upon productivity than capital intensity on its own (Koch and McGrath 1996). Hiring lead professionals from frontline firms, also results in knowledge transfers that contribute to the hiring firm's TFP, but only for firms with high absorptive capacity (Lodefalk and Tang 2018).

The productivity of the firm's workforce is also contingent on factors such as work systems, business awards, industrial relations, and technological innovations (Antonioli et al. 2010; Fabling and Grimes 2014; Henriques and Catarino 2016; Jones et al. 2014; Mayer et al. 2017; Towers and McLoughlin 2005). High-performance and decentralized incentive-based work systems are generally positively related to labour productivity. (Datta et al. 2005; Guthrie 2001; Huselid 1995; Rasel 2016), yet it is still unclear whether high performance systems lead to increased productivity or vice versa (Datta et al. 2005). Moreover, the negative effect that small firm size can have on the combination of decentralized work systems and technological innovations points to the need to better

understand how workplace systems interact with other internal factors to impact the labour productivity of different sized firms (Rasel 2016). Unlike large firms, which have more comprehensive work management, HR practices, and technology systems, SMEs tend to focus on organizational practices that increase revenues and optimise resources in order to improve their level of productivity (Bartel et al. 2007; Torrent-Sellens et al. 2016).

Given the increased adoption of technology by SMEs, examining how workplace practices might interact with technological innovations to influence firm productivity is another central focus of reviewed studies (Cohen, 2006; Colombo et al., 2013; Fabi et al., 2010; Raymond, 2005; Torre and Solari, 2011). These point to a complementary relationship, with extensive strategic and organizational changes required to accompany the adoption of technological innovations - and vice versa - in order to have a positive impact on productivity (Colombo et al. 2013; Torre and Solari 2011). Studies analysing the link between productivity, innovation strategies, and industrial relations, find that cooperative industrial relations positively impact training activities and organizational innovation, which in turn drive superior labour productivity (Antonioli et al. 2010). Additionally, Black and Lynch (1996) find that both industrial relations and recruitment processes focused on grades have a positive impact on the labour productivity of firms in the non-manufacturing sector.

Further, studies analysing firm ownership structure focused on understanding how the unique characteristics of family firms, such as family members involvement in the daily running of the firm or the desire to transfer ownership through succession, might influence a firm's productivity level (Barbera and Moores 2013; Cassia et al. 2012; Classen et al. 2014). While no major differences are found between family controlled firms and non-family firms in terms of investment in innovation or financial performance, the former tend to be disadvantaged in terms of workforce motivation and labour productivity (Cassia et al. 2012; Classen et al. 2014). However, these findings seem to be influenced by the productivity measure used. While the two previous studies focus on measuring the labour productivity of firms in Italy and Germany respectively, a study by Barbera and Moores (2013) focuses on measuring the TFP of firms in Australia. This latter study finds that family-controlled firms' labour output (capital output) contributions are significantly higher (lower) than that for non-family firms. Subsequently, they conclude that when heterogeneity of output contributions are considered, no differences exist between family controlled and non-family firms' productivity levels (Barbera and Moores 2013).

Although numerous studies have been carried out on understanding the impact of a firm's internal organizational environment on productivity, there is still scope for future research to analyse the interrelationship between ownership structure and factors like technological innovations, R&D, and HR systems. Studies examining a wider range of workplace practices, and those considering the intensity of technology usage, as opposed to mere adoption, could also provide a more in-depth understanding of the conditions under which these relationships are magnified (Colombo et al. 2013). Moreover, the absence of longitudinal studies in this area limits our knowledge of the cumulative impact that HR practices or training schemes might have on SME productivity.

Theme II – Organizational capabilities: the studies under this theme focused on understanding how a firm's ability to manage its human resource impacts its level of productivity. Studies analysed a variety of factors including staff employability, training, wellbeing, leadership, and owner-managers' demographics (Arocena et al. 2007;

Cocker et al. 2013; Henriques and Catarino 2016; Li et al. 2016; McGuirk et al. 2015; Middleton and Byus 2011; Ribeiro Soriano and Castrogiovanni 2012; Savery and Alan Luks 2004). In general, training seems to have a positive effect on labour productivity (Georgiadis and Pitelis 2012; Mancinelli and Mazzanti 2009). However, a more in-depth analysis reveals that the impact of training activities is dependent on other factors. Training programmes focused on employees have a larger positive impact on firm productivity than those focused on managerial staff or HRM (Georgiadis and Pitelis 2012). Further, Black and Lynch (1996) find that the average education level of firm employees has a significant positive effect on labour productivity for both the manufacturing and non-manufacturing sectors. Moreover, they also find that while a greater proportion of formal off-the-job training resulted in higher labour productivity (after adjusting for costs of introducing new skills) for firms in the manufacturing sector, the content of training programs was more important for those in the non-manufacturing sector (Black and Lynch 1996; Koch and McGrath 1996). The need to match employees' skill levels to the firm's level of technological change is also highlighted by Sandulli et al. (2013) who find that, contrary to common belief, having a younger workforce could result in reduced efficiency for firms with high-levels of technology change. Highly skilled employees are also found to be as equally important as R&D for the innovation output of knowledge-intensive service micro-firms (Audretsch et al. 2020).

Additionally, the slow adoption of information and communication technologies (ICTs) by SMEs has been shown to constrain productivity growth (Martínez-Caro and Cegarra-Navarro 2010). A number of studies analyse the relationship between SME owner-managers' socio-demographic characteristics (e.g. gender, ethnicity, race, education, experience) and their adoption of ICTs (Middleton and Byus, 2011; Middleton and Chambers, 2010). Whereas no significant effect is found for the owner-managers' gender, education, or experience, on their intention to adopt or use Wi-Fi technology, this is not the case when ethnicity is considered (Middleton and Chambers, 2010). Examining SMEs located in renewal communities - areas specifically targeted for growth and development by the federal government in the USA - Middleton and Byus (2011) find a continuing digital divide between Hispanic and non-Hispanic SME-owners, which undermines the very policies designed to enhance business operations in these relatively disadvantaged urban areas. In general, non-Hispanic SME-owners are more likely to adopt ICTs for both administrative and analytical purposes (Middleton and Byus, 2011). Interestingly, another study analysing the impact of a workforce's cultural diversity on patent applications in various German regions, finds that cultural diversity has an innovation-enhancing effect due to its positive influence on the knowledge creation process (Niebuhr 2010).

Further, only a handful of studies focus on leadership and employee wellbeing, even though these factors play a key role in influencing productivity. For example, Ribeiro Soriano and Castrogiovanni (2012) find that the CEO's entrepreneurial human capital, in the form of industry specific knowledge prior to start up, and general business knowledge post-start-up, are both positively related to the firm's labour productivity. Furthermore, Li et al. (2016), who focus on the impact of e-leadership, find that SME leaders play an important role in improving firm productivity by ensuring optimal utilization of existing/new technologies. SME owner-managers' wellbeing also has an important impact on labour productivity, with experiences of high levels of psychological distress resulting in substantial productivity loss (Cocker et al. 2013). However, as these two topics remain relatively

underexplored, future studies examining the interrelationship between employee wellbeing, leadership styles, and SME productivity can make significant contributions to current knowledge. Moreover, there is need for a better understanding of the type of training that is most appropriate for increasing labour productivity; especially when factors such as the level of education/experience of the workforce, or the sector/region within which the firm operates, are taken into consideration.

Theme III – Investments: the studies classified under this theme analysed the impact of investments in R&D and technology on firm productivity. While the relationship between technology and productivity has been a source of debate historically (Brynjolfsson and Yang 1996), recent studies point to the positive impact intensive use of information technology has on productivity. This is attributed to the increased collaboration capabilities (Ruivo et al. 2015), as well as the decreased market research and coordination costs that exporting firms incur (Sandulli et al., 2012). Despite the fact that investment in technology is shown to be relevant for SMEs, both in terms of productivity and firm survival in competitive markets (Niaki and Nonino 2017; Ruivo et al. 2015; Scuotto et al. 2017), SMEs are still slow in adopting ICTs (Martínez-Caro and Cegarra-Navarro, 2010; Mole et al., 2004). Two of the studies in our sample analyse how owner-managers' perceptions might influence SMEs' adoption of new technologies (Nunes et al. 2006; Wielicki and Arendt 2010). Nunes et al (2006), for example, find that while owner-managers are cognizant of the positive impact of knowledge management on productivity, they are still reluctant to invest in long-term goals for which they cannot establish added value. Thus, knowledge management activities within SMEs happen in an informal manner, and are rarely supported by ICT systems (Nunes et al., 2006).

Moreover, an SME's ability to assimilate technological innovations, rather than simple adoption, is a crucial factor in determining whether it experiences a positive or negative impact on its productivity levels (Raymond and St-Pierre 2005). R&D is crucial to facilitating a firm's absorption capacity of new technology, and through innovation, boosts productivity growth (Ballestar et al. 2020; Parisi et al. 2006). Internal R&D and knowledge spillovers are also complementary for labour productivity; with knowledge spillovers being more important for productivity, and R&D being important for both innovation and productivity (Audretsch and Belitski 2020). Knowledge spillovers also have an impact on the intensity of R&D investment, which subsequently results in increased labour productivity growth (Audretsch and Belitski 2020). Analysing robotic tool adoption in Spanish manufacturing SMEs, Ballestar et al. (2020) also find that while knowledge spillover effects relate to capitalization and exports in the case of robotized SMEs, they relate to human capital for non-robotized SMEs.

Studies also focus on understanding how public support might influence the likelihood of a firm's engagement in R&D and the subsequent impact on productivity, but with inconclusive results. On the one hand, some studies find that public subsidies have a positive impact on a firm's investments in R&D (Griffith et al. 2006; Hottenrott and Lopes-Bento 2014). In this case, increased R&D efforts per employee leads to a higher probability of the firm having at least one process or product innovation; with both types of innovation being associated with higher firm productivity (Griffith et al. 2006). Publicly induced R&D investments also positively impact sales from market novelties, especially for international collaborators, suggesting that knowledge spillovers from partner firms contribute substantially to a firm's success when introducing radical innovations (Hottenrott and Lopes-Bento

2014). On the other hand, other studies find that R&D subsidies have no positive significant effect on labour productivity over the five-year period after a subsidy is granted (Karhunen and Huovari 2015). In effect, due to its negative impact on productivity, small firms tend to invest in R&D only when public support is available (Serban 2013).

While reviewed studies highlight the varied impact that investment in technology and R&D have on productivity based on factors such as firm size, age, and sector (Baumann and Kritikos 2016; Nunes et al. 2013; Struker and Gille 2010), there is still a limited understanding of the lagged effect of R&D spending, or the impact of the economic cycle, on SME productivity growth (Madrid-Guijarro et al., 2013; Mañez et al., 2013; Sandulli et al., 2013). Further longitudinal studies that are better able to capture the dynamic aspects of such factors would make valuable contributions to this knowledge gap. Future research should also consider the interrelationship between technology, R&D investments, organization culture, and management style (Falk and de Lemos, 2019; Nunes et al., 2006).

Theme IV – Types of innovation: the studies under this theme focused on understanding how a firm's participation in either process or product innovation, or a combination of the two, might influence productivity. Although the studies highlight that both types of innovation have a positive impact on firm productivity (Bagchi-Sen 2001; Dora et al. 2014; Hall et al. 2009; Higón and Driffield 2011; Maani et al. 1994; Raymond et al. 2013; Su and Tang 2016), a firm's decision to undertake either type of innovation is influenced by the firm characteristics (Calderini and Cantamessa 1997; Dora et al. 2013; Su and Tang 2016) and level of R&D spending (Baumann and Kritikos 2016; Griffith et al. 2006; Hall et al. 2009; Parisi et al. 2006).

Examining the link between R&D, innovation, and labour productivity for micro- and large SMEs in Germany, Baumann and Kritikos (2016) find that only product innovation has a sizable positive effect on labour productivity for both micro-firms and SMEs. Labour productivity also increases with firm age, and particularly in the case of micro-firms (Baumann and Kritikos 2016). Similarly, Su and Tang (2016) who analyse the factors influencing a firm's decision to pursue different business strategies in a post-crisis context, find that Canadian SMEs pursuing product innovation are more productive than those pursuing cost-cutting strategies. On the contrary, while investigating how and when innovation takes place, Hall et al. (2009) find that both product and process innovations (and especially process innovation) have a positive impact on the productivity of SMEs, and that larger and older firms seem to be, to a certain extent, less productive, ceteris paribus. Similarly, Audretsch et al. (2020) find that start-ups and young knowledge-intensive service micro-firms are more likely to engage in innovation activities than mature firms, and that such innovation activities lead to increased labour productivity.

The lagged effect that process innovations might have on TFP is also analysed in a study by Manez et al. (2013), who find that implementing a process innovation for the first time does not guarantee immediate productivity rewards. Instead, it allows the firm to gain experience which results in extra productivity growth in the subsequent period, and that this growth lasts for only one period (Mañez et al. 2013). A more holistic approach that incorporates organization-wide cultural changes is required in order for a firm to benefit from productivity gains over the long-term (Gunasekaran et al. 2000; Uwizeyemungu et al. 2015). Therefore, a more comprehensive

understanding of the lagged effect of product and process innovations on SME productivity, as well as the dynamic relationship between innovation processes and IT systems integration (Raymond et al., 2013) is needed. Greater attention should also be paid to the type of firm innovation (i.e. incremental versus radical) and its impact on productivity (Hernandez-Espallardo and Delgado-Ballester 2009).

Theme V – External knowledgebase: this theme includes studies analysing the impact of alliances/collaborations, business advisors, takeovers or firm mobility, clusters and value chains on SME productivity. Most studies in this theme point to the positive impact that alliances/collaborations, value chains, and knowledge clusters have on SMEs' innovation capabilities and productivity (Antonelli et al. 2015; Bakhtiari 2015; Capello 1999; Hemert et al. 2013; Jones and Corral 2017; Noke and Hughes 2010; Tomlinson and Fai 2013). Specifically, the formation of alliances/collaborations is associated with increasing firms' new product capability (Noke and Hughes 2010), facilitating the ICT adoption process for micro firms (Gare and Melin 2011), and increasing technological capabilities (Barajas et al. 2016); all factors that are associated with improved labour productivity. Audretsch and Belitski (2020) also find that increased investment in R&D and knowledge spillovers impact the likelihood of an owner-manager implementing "make or ally" innovation strategies rather than "buy or imitate" new products developed by others.

Furthermore, participation in publicly supported international research joint ventures or knowledge transfer partnerships at local/international levels also allows SMEs to draw upon higher institutions' knowledge and human resources to strengthen their innovation capabilities and increase productivity (Barajas et al. 2016; Humphries 2005). Nevertheless, the positive impact of such collaborations is contingent upon firm specific factors. For example, Mancinelli and Mazzanti (2009) find that without accompanying investments in R&D and training, networking (and by implication the mere existence of local spillovers) plays a negligible role in stimulating labour productivity. In the same way, the impact of productivity centres on SMEs' process improvement capabilities are dependent on: firms' willingness to change, senior management support, establishment of key performance indicators, and availability of personnel (McGovern et al., 2017). SMEs therefore need to make internal organizational adjustments in order to reap the benefits of increased productivity resulting from participation in external collaborations and networks.

Other aspects examined by studies in this theme are the role of advisors, and the relationship between takeovers/firm mobility and productivity. Henley and Song (2019), for exmaple, find that using external advice, business networking, and having a formal business plan, are all associated with a significant increase in the likelihood that micro-firms will innovate. Studies focused on analysing interventions aimed at increasing SMEs' innovation capabilities also find that, while such interventions have a positive impact on goal setting, firm performance, and labour productivity, they have a detrimental effect on innovation activities (Oeij et al. 2015). For example, Harris et al. (2016) find that SMEs that are located in peripheral regions and that adopt business improvement methods end up undertaking innovation-related activities but remain non-innovators.

A number of studies also find a reciprocal relationship between SMEs takeover or mobility and the level of productivity, that is further influenced by geographic location (Foreman-Peck and Nicholls 2013, 2015). Unlike

large firms, more productive small firms are at a greater risk of takeovers if they are centrally located (Foreman-Peck and Nicholls 2013). Such takeovers raise the productivity levels for SMEs located in both central and peripheral regions. Moreover, firm mobility is more prevalent amongst SMEs that are younger, more productive, relatively larger, and centrally located, than regionally immobile firms (Foreman-Peck and Nicholls, 2015). After moving, these fast-growing SMEs become even more productive and employ more workers; and through this market process the regional core supports peripheral regions (Foreman-Peck and Nicholls 2015).

Although the studies highlight the important role that the external knowledgebase plays in influencing SME productivity, a number of issues remain unresolved. For example, it is not clear whether cooperation leads to innovation or vice versa (Tomlinson and Fai 2013). The role of firm specific factors such as ownership structure, firm size or age, in influencing the formation of external collaborations, as well as the subsequent impact on innovation and productivity is equally unclear (Harris et al. 2016; Noke and Hughes 2010). Therefore, further indepth analyses of these relationships would make valuable contributions in this area.

Theme VI – Commercialization: studies under this theme focus on analysing how global competition, export activities, and the external environment influence SME productivity (Guzman-Cuevas et al. 2009; St-Pierre and Raymond 2004). The strong positive relationship found between SMEs' productivity levels and export participation (Eliasson et al. 2012; Falk and de Lemos 2019; Falk and Hagsten 2015a, 2015b) has resulted in research focused on understanding the factors influencing an SME's intention to export. A number of studies show that factors such as innovation activity, firm size, workforce, and public incentives all influence a firm's propensity to export (Banno et al. 2014; Eliasson et al. 2012; Falk and de Lemos 2019; Higón and Driffield 2011; Onkelinx et al. 2016). Henley and Song (2019), however, find that there is no direct productivity benefit from innovation for micro-firms, but that the impact is as a result of exporting as an internationalisation strategy. Importantly, the intention to export also has a positive impact on a firm's productivity levels (Eliasson et al. 2012).

Further, investigating the export productivity premium of SMEs, Falk and Hagsten (2015b) find that the level of labour productivity of exporting SMEs is 13 percentage points higher than that of non-exporting firms in any given industry and country. Relative productivity levels are also higher for SMEs present in both EU and non-EU markets than those operating in only one of the two markets. A second study by Falk and de Lemos (2019) finds that SMEs' export participation and export share depend on their labour productivity level and R&D-sales ratio, with the impact being more pronounced for micro-, young, and born global firms. Firm labour productivity (Onkelinx et al. 2016) and industry sector (Lejarraga and Oberhofer 2015) also have an impact on SME internationalization and intention to export respectively, with productive ICT and professional service firms more likely to export to foreign markets (Lejarraga and Oberhofer 2015). Additionally, Lejarraga and Oberhofer (2015) show that TFP and firm size of SMEs in the manufacturing and services sector affect their probability to export, but not their relative engagement in foreign markets. Public financial support is also found to be effective in enhancing productivity growth, especially when targeted towards smaller, younger firms, and those with international experience (Banno et al. 2014).

Other studies examining the potential contribution of SMEs to rural economic development find that rural regions have a higher proportion of fast growing SMEs, which are more active in adapting to local market conditions (North and Smallbone, 1996). However, these SMEs lag behind their urban counterparts, both in terms of labour productivity, and orientation to national and international markets (North and Smallbone 1996). A more recent study analysing productivity transmissions between SMEs and large firms, confirms a positive spillover effect of SME labour productivity on large firm productivity in the long run (van Stel et al. 2019). Similar positive spillover effects on TFP are experienced between frontier and laggard firms (Romero-Jordan et al. 2019). Finally, labour productivity is an important determinant for growth and profitability of SMEs in various industry sectors, with smaller, younger firms benefitting more than larger, older firms (Guedes de Carvalho et al. 2013; Yazdanfar 2013).

Given the significant role that global competition and participation in exports play in enhancing SME productivity, future studies examining how related factors such as location of global markets, or human capital (e.g. level of education or international work experience) might influence a firm's export intentions (Kohr et al. 2017; Onkelinx et al. 2016) can provide further insights in this area.

4 Discussion and Future Research Agenda

SME productivity growth is important for developed economies as they adapt to major trends such as digitalisation, an ageing population, and changes in the nature of work. For example, the BEIS report on small businesses and productivity indicates that SMEs play an important role in contributing to productivity growth through their impact on wealth and employment creation (BEIS 2018b). Fostering productivity in SMEs as a way of ensuring their survival and growth has therefore been a major focus for both practitioners and policymakers (Mañez et al. 2013). The continuing decline in labour productivity growth that characterized many western economies since mid-2000, and especially after the financial crisis of 2008-2009, has similarly led to a sharp increase in SME productivity research. However, as underlined by our systematic review, which aimed to provide an integrated understanding of the main factors affecting SME productivity, this research field is highly fragmented. While the empirical studies provide useful insights into the direct, indirect, and mediating/moderating factors influencing SME productivity, there are important gaps in knowledge that need to be addressed by future research. Our discussion highlights four main areas, namely 1) the need for comparative analyses within SMEs, as well as between SME and large firms, 2) the need to adopt a holistic approach to understanding human capital in SMEs, 3) the importance of context, and 4) the methodological challenges and developments to be considered. In highlighting these gaps, we develop a research agenda to guide future research and help advance knowledge on SME productivity. We also consider the limitations of our study at the end of this section.

4.1 The need for comparative analyses

Our analysis emphasizes the effect of firm characteristics (e.g. size, age, sector) on productivity levels (Barbera and Moores, 2013; Eliasson et al., 2012; Falk and Hagsten, 2015a; Hall et al., 2009; Laforet, 2013; Mole et al.,

2004; Raymond et al., 2013; Sandulli et al., 2012). Indeed, firm size can have a constraining effect on firm capacity both in terms of its human and financial resources, and influences the type of strategies SMEs adopt to improve productivity levels (Torrent-Sellens et al. 2016). Besides, a combination of firm size, age, and sector, influences a firm's innovation capability and determines the effect that technological and R&D investments have on its productivity (Baumann and Kritikos, 2016; Hall et al., 2009; Nunes et al., 2013). Apart from having an impact on the internal firm factors influencing productivity, these firm specific characteristics can also act as enablers or inhibitors to external factors such as the types of alliances/networks formed, or level of participation in global markets (Harris et al. 2016; Noke and Hughes 2010). For example, network relationships with suppliers, customers, trade associations, and other intermediaries, are important factors affecting innovation and SME productivity (Pittaway et al. 2004).

A firm's sector will also influence the type of opportunities and restrictions impacting its growth and productivity level. For example, Hogan and Coote (2014) highlight that a culture of innovation in professional services firms in particular, supports innovative behaviours that can sustain organizations and foster organizational renewal. Even the type of training most appropriate for increasing SME labour productivity is influenced by the sector in which the SME operates (Black and Lynch 1996). Future studies therefore need to pay greater attention to changing trends in different sectors. The growing economic importance of the services sector is likely to result in R&D activities becoming more prevalent in order to support service innovations (Miles, 2007), with potential benefits for productivity enhancements.

Therefore, in line with Mallett et al. (2018), we find the homogenisation of different-sized businesses under the SME umbrella problematic, as it obscures differences that may be important for understanding firm operations, goals, and challenges. For example, in the case of micro-firms "the owner-manager is the company" (Lean, 1998:233), which not only imposes constraints on their growth capacity but also makes growth dependent on the owner-managers' skills, capabilities, and ambitions (Gherhes et al. 2016). In such firms, owner-managers have a more dominant role than might be the case for larger counterparts, which can have key implications for productivity. Future studies on SME productivity therefore need to consider firm size and the heterogeneity of firms are classified under this category, especially given that definitions of SMEs tend to vary across geographic regions (Koch and McGrath 1996; Raymond et al. 2013; Raymond and Bergeron 2008). This would allow for more in-depth analyses into the types of factors that are more relevant for specific firm categories, while providing greater insights for targeted interventions aimed at improving SME productivity.

Furthermore, there is a need for more comparisons between SMEs and large firms in order to better understand the type of factors that have a greater impact on the productivity of different sized firms. For example, exporting, like many other factors highlighted in this review is not restricted to SMEs, but affects firms of all sizes. While studies find an export premium, with export entrants becoming more productive once they start exporting (De Loecker 2007, 2013), key questions that remain unresolved include whether export premium varies with firm size. Future research could focus on both within-SMEs variation, as well as compare SMEs to large firms, in order to understand the benefits of exporting on productivity gains.

4.2 Human capital in SMEs

The studies in our review indicate that human capital is significant in determining an SME's level of productivity. However, this area remains relatively underexplored, with only a handful of studies looking at aspects such as leadership style, organizational culture, and employee wellbeing (Cocker et al. 2013; Li et al. 2016; Ribeiro Soriano and Castrogiovanni 2012). Importantly, reports indicate that poor management practices can have a limiting effect on SME productivity and innovation (OECD 2017b). CEOs' managerial skills and human capital, especially in the case of SMEs, are important in promoting innovations within the organization (Custodio et al. 2017). The current gap in knowledge therefore presents opportunities for future studies in this area. Given that management practices account for 30% of TFP differences both within countries, and between countries across firms (Bloom et al. 2016), we encourage future studies to examine how the CEO's management and/or leadership style directly or indirectly influences SME productivity (Büschgens et al. 2013; Hogan and Coote 2014).

Moreover, a more holistic approach to analysing human resources in SMEs that focuses not only on skills and experience, but also on factors such as workforce diversity, wellbeing, and work-life balance, is necessary. Given the positive impact that ethnic and gender diversity has on firm innovation (Díaz-García et al. 2013; Nathan 2014), future studies analysing these and other types of diversity (e.g. neurodiversity, ability status, race) would provide insights on the relationship between diversity and productivity. Understanding how modern societal challenges, such as an ageing population and changes in the nature of work (e.g. more flexible and/or short-term contracts) might impact productivity also provides a fruitful avenue for future research.

Our knowledge on the impact of wellbeing and work-life balance on productivity, or vice versa, also remains limited. Work-life programmes, for example, have a positive impact on productivity in large firms (Konrad and Mangel 2000). However, given the different operational dynamics that SMEs are likely to face, future studies could examine how work-life balance is achieved within SMEs, and how this subsequently impacts productivity. Wellbeing has also been strongly associated with active engagement in entrepreneurial tasks (Shir et al. 2018). This research should be extended to determine whether similar positive relationships exist between wellbeing and SME productivity. We also argue that the different aspects of human capital should not be analysed in isolation. Instead, studies should focus on understanding how they interact with other internal firm-level factors, such as investment in R&D, technological innovations, workplace systems, as well as external firm factors like the external knowledgebase to influence productivity.

4.3 The importance of context

In management research, and for the purposes of this systematic review, context refers to the circumstances, conditions, situations, or environments that are external to the respective phenomenon and that enable or constrain it (Welter 2011). Critically, only one study in our sample focused on analysing the impact of socio-economic contexts on SME productivity, which was insufficient to warrant a separate theme. The study examined how spatial economic and socio-institutional factors might influence the productivity of SMEs located in the Northern and Southern provinces of Italy (Fazio and Piacentino 2010), with the socio-economic context being measured in

terms of: level of employment, unemployment rates, crime levels, and net brain drain. They find that even though SMEs located in regions of greater social decay experience lower productivity, this relationship is dependent upon the firm's level of capital intensity. Hence, firms with higher capital intensity are less affected by external factors such as location or geography (Fazio and Piacentino 2010).

We therefore want to highlight the important role of the socio-economic context for SME productivity levels, especially in the context of broader calls for contextualizing entrepreneurship research (Autio et al. 2014; Smallbone 2016; Welter 2011). Scholars have shown that a firm's geographic location matters (Foreman-Peck and Nicholls, 2013; Foreman-Peck and Nicholls, 2015; Harris et al., 2016; North and Smallbone, 1996), with SME productivity being influenced by external factors at the national, regional and local levels. At the regional and local level, external factors such as the availability and cost of labour, market size, level of infrastructure, technology, as well as access to finance and knowledge centres, can impact SME productivity. At the national level, formal (legislations, regulation, property rights) and informal (social norms, cultures, customs, traditions) institutions, also referred to as the rules of the game (North, 1991) provide the incentive structure for economic activity. Institutions can therefore act as either enablers or inhibitors of entrepreneurial behaviour (Welter and Smallbone 2011), firm productivity (Ostapenko 2015), and innovation (Autio et al. 2014). However, studies focused on cross-country or regional level comparisons are scarce, thus limiting our understanding of the dynamics and uniqueness of different country and regional contexts (Autio et al. 2014; Smallbone 2016; Welter 2011), as well as the generalisability of current knowledge across different contexts. The lack of multi-country comparisons also provides opportunities for future research aimed at gaining a deeper understanding of the national/regional factors influencing SME productivity. We therefore call for future context-sensitive research to provide a better understanding of how different contextual factors enable or constrain SME productivity.

Further contributions can also be made by studies analysing the direct and indirect impact of informal institutions (i.e. culture, social norms, societal expectations) on SME productivity. For example, the intersection of gender and innovation leads to gendered innovation processes, and the gendering of innovation practices (Alsos et al. 2013). Socio-economic factors also tend to have a constraining influence, positioning female-led firms in gendered spaces (Carter et al. 2015), and impacting the recognition of opportunities for social innovations (Spiegler and Halberstadt 2018). Analyses of female SME owner-manager's innovation activities and the resulting impact on productivity, should therefore be embedded in understanding normative frames and structural factors at play (Alsos et al. 2013). Future research examining the relationship between factors such as culture, ethnicity, gender, race, and productivity can provide useful insights on the broader structural challenges that SME owner-managers belonging to under-represented groups encounter, and the strategies they employ to overcome them.

4.4 Methodological challenges and developments

As productivity measurement is strongly data-driven, access to and quality of firm-level data is particularly important. A key challenge that we identified in this systematic review is the variety and inconsistency of methodological approaches adopted, with a wide range of data sources, sampling methods, and data analysis

techniques being employed to research SME productivity. As such, there is a need to consider the advantages and disadvantages of specific data. For example, self-conducted surveys of production values are likely to suffer from measurement errors and high rates of non-response. Balance sheet data is also generally not appropriate for analysing SME productivity, due to reporting thresholds that result in micro- and small firms not providing profit and loss accounts in many countries. Another important challenge is firm-level data representativeness; given that larger SMEs are often over-represented while micro firms are under-represented.

Main issues that require greater consideration by future research therefore includes the low response rates of individual surveys, unclear representativeness, gaining access to national statistics data, and linking data from different surveys. For instance, while data from national statistical offices is of high quality, access and linking possibilities are country specific. Therefore, there is a need for multi-country studies using harmonised and internationally comparable data at firm level (Hagsten and Kotnik 2017). Moreover, the type of data available typically includes either administrative accounting data with little opportunity to link to contextual information, or surveys focused on contextual factors with no opportunity to link to value-added firm-level data. In the latter case, the answers to productivity questions could also depend on how respondents frame their understanding of productivity. Nevertheless, data linkage remains a key issue, which also presents the problem of connecting firm-level data to information on the characteristics and attributes of individuals, such as owner-manager teams rather than single individuals.

Furthermore, while studies use both labour productivity and TFP measures indiscriminately, we argue that greater attention needs to be paid to the type of measurement used, as this might influence study results (Barbera and Moores 2013; Cassia et al. 2012; Classen et al. 2014). For example, Linna et al. (2010) highlight the limitations of using single factor productivity measures, such as labour productivity, that attribute all increases of efficiency to one factor. Moreover, while TFP measures might be more appropriate in determining SME productivity given the changing trends in the work environment, future studies need to bear in mind the various challenges in estimating TFP (Van Beveren 2012).

The issue of endogeneity is another key concern for future econometric studies on SME productivity. Our systematic review indicates that only 31% of quantitative studies deal with this issue, and these employ various methods such as lagged or predicted variables, Hausmann-Wu test, CDM framework, or the Generalised Method of Moments system estimator. As most explanatory variables are correlated with the error term, and given the restrictiveness of OLS Gauss-Markov assumptions, coefficients may be biased and/or inconsistent if the issue of endogeneity is not adequately addressed. At the same time, given that 85% of the reviewed studies are quantitative, there is not only scope, but also need for more mixed-method research to untangle the complex nature of SME productivity.

Finally, the analysis of issues mentioned in the previous sections also calls for research using a wide array of methodological approaches and data sets. For example, a more fine-grained analysis of the contextual factors influencing SME productivity might provide opportunities for studies using qualitative or mixed method approaches. Additionally, the lagged effect of R&D investments, the long-term cumulative effects of interventions

such as training programmes, the adoption of technological innovations, and the changing work environment, all call for longitudinal studies that are better able to capture the changing dynamics of SME productivity growth. Experiments and randomised controlled trials—not widely used in the SME literature—would also allow studies to move beyond cross-sectional analyses to determine causal relationships between factors such as organizational culture, innovation, HRM capabilities and SME productivity growth (Baumann and Kritikos 2016; Fabi et al. 2010; Naranjo Valencia et al. 2010).

4.5 Limitations

We acknowledge a number of limitations to our systematic review. First, while the systematic review methodology is designed to be as objective as possible, by its nature, it relies on defined parameters, which means that certain papers will be out of scope, but not irrelevant. Thus, the use of keywords, search strings, and specific databases may have limited the number of papers identified as relevant by the review, while the databases used may also exclude publications from specific journals, making it difficult to capture all relevant literature. Second, restricting database searches to titles and abstracts may also have restricted the number of articles found. However, such an approach is deemed appropriate when the body of knowledge is vast (Thorpe et al. 2005), which is the case of the SME productivity literature. Third, the high variation in data sources, research methods, and data analysis techniques in our sample of articles hinders the direct comparability of study findings. Nevertheless, our systematic review provides a useful map of the SME productivity literature, highlighting key methodological shortcomings and providing key directions for future research in this area.

Therefore, we acknowledge that certain key issues that are relevant to the SME productivity debate may not have been captured by our systematic review due to the nature of the review protocol. For example, the issue of knowledge spillover effects that have been shown to have an impact on innovation and productivity growth (Acs et al. 2009; Glaeser et al. 1992) is relatively underexplored by the reviewed studies. Specifically, only 14% of the reviewed studies address the issue of spillover effects, and examine, for example, the spillover effects of SME productivity to large firms (van Stel et al. 2019), spillover effects of TFP growth from frontier to laggard firms(Romero-Jordan et al. 2019), the relationship between internal R&D, knowledge spillovers and innovation strategies (Audretsch and Belitski 2020), and the link between knowledge flows, industrial robotics, and labour productivity (Ballestar et al. 2020). The remaining studies focus on analysing the impact of alliances, collaborations, recruitment practices, or workforce diversity on firm productivity (Hottenrott and Lopes-Bento 2014; Lodefalk and Tang 2018; Niebuhr 2010), and indicate possible knowledge spillover effects as part of their findings.

Similarly, access to capital has been shown to have major implications for SME productivity. A meta-analysis carried out by Kersten et al. (2017) on SMEs in low and middle income countries, finds a positive significant effect of SME finance on investments, firm performance, and employment. Although, the reviewed studies do acknowledge the particular challenges faced by SMEs in accessing finance, the issue of funding is primarily analysed in relation to the impact that public funding and subsidies for R&D investment, internationalization, or

research joint ventures, have on SME productivity (Banno et al. 2014; Barajas et al. 2016; Griffith et al. 2006; Hottenrott and Lopes-Bento 2014; Karhunen and Huovari 2015). While our systematic review may not have captured a wide range of papers focused on these topics, there is scope for future reviews to provide additional insights in these and other areas, and thus to bring further clarity to this highly fragmented field.

5 Policy Implications

Our systematic review yields important implications for policy. Improving productivity as a means of ensuring long-term economic growth is a key concern for many governments, with SMEs being central to achieving this growth (van Stel et al. 2019). For example, the productivity lag in the UK has been attributed to the long tail of unproductive small business (BEIS 2018b). Gaining a better understanding of the main factors affecting productivity in SMEs is therefore crucial for researchers and policymakers alike (Mañez et al. 2013). As emphasised by the OECD, "upgrading productivity in a large population of small businesses, including in traditional segments and the informal economy, can help governments achieve both economic growth and social inclusion objectives, including escaping from low productivity traps and improving the quality of jobs for lowskilled workers" (OECD 2017b, p.8). Implications for policymakers include the need to provide sustainable support to SMEs, which focuses on building skills and competencies that can assist them in implementing organizational/strategic changes needed to profit from sustained productivity gains as a result of R&D and technological investments (Gunasekaran et al. 2000). Targeted interventions such as training and/or advisory programmes should also consider the mode and type of training being offered, as well as firm specific factors such as size, age, sector, and location. Providing a platform that facilitates greater collaboration and networking opportunities between SMEs and local/international knowledge centres (e.g. universities, research institutes) should also have a positive effect in strengthening SMEs' innovation capabilities. Lastly, interventions that are focused on providing public subsidies as a way of countering the negative effects of R&D investment should also be sensitive to unintended consequences, such as the survival of less efficient firms (Karhunen and Huovari 2015), that could subsequently have a negative impact on overall productivity growth.

In this sense, the issues highlighted in our systematic review relate to the broader productivity puzzle debate. This refers to the unusually weak or declining productivity growth that has characterized many western economies since the financial crisis of 2007-2008 (OECD 2017c). While competing explanations have been proposed for this declining growth, such as the difficulty of measuring productivity, or the shortage of demand and investment opportunities that constrain productivity growth, economists have hitherto failed to reach a consensus. However, it is widely acknowledged that productivity remains the single most important economic factor (Prowle et al. 2017). As Paul Krugman's widely cited remark emphasizes: "Productivity isn't everything, but, in the long run, it is almost everything." A country's ability to improve its standard of living over time depends almost entirely on its ability to raise its output per worker. Importantly, SMEs make a critical contribution to economic growth and job creation in western economies, yet differences in productivity between large and small firms persist, with SMEs often characterised by lower productivity than their larger counterparts (OECD 2018). In highlighting methodological shortcomings and key gaps in our knowledge on SME productivity, our systematic review

provides a useful map for future research to address these issues, and provide critical insights into SME productivity dimensions that can contribute towards solving the productivity puzzle.

6 Conclusion

Our systematic review of 109 empirical studies aims to provide an integrated understanding of the main factors affecting SME productivity in western economies. By thematically analysing studies, we provide a map of the current SME productivity research landscape, identifying direct, indirect, and mediating/moderating factors influencing SME productivity. Our systematic review highlights the fragmented nature of research in this field, and makes three key contributions. First, we thematically map existing empirical studies into six key themes organised around internal (organizational environment, organizational capabilities, investments and types of innovation) and external (external knowledgebase, commercialization) factors influencing SME productivity. Second, our descriptive and thematic analyses enable us to highlight critical gaps in existing knowledge, as well as methodological issues that restrict our understanding of how different factors such as technology, R&D, leadership skills, networks and collaborations etc., interact to affect SME productivity. Third, we develop a research agenda and propose four main areas where future research could make useful contributions in advancing knowledge on SME productivity. These include the need for greater comparative analyses (i.e. within SMEs as well as between SMEs and large firms), the need to adopt a holistic approach to understanding human capital in SMEs, the importance of context, and methodological developments that would facilitate the analysis of underlying factors affecting productivity.

Future studies can build on our systematic review by focusing on SME productivity in emerging or developing economies, in order to provide a better understanding of the different macro-level factors driving firm productivity in these contexts (Rangamohan et al. 2007). Future reviews could also focus on countries with robust firm-level data such as Japan, South Korea, and China, that have been excluded from the current study due to the parameters of the systematic review. Moreover, there is scope and need for meta-analyses that draw on results from extant studies to provide better clarity on the influence of specific factors on SME productivity, especially in cases where results are inconclusive. These can also address the some of the methodological challenges highlighted in our discussion. Finally, given the importance of the topic for practitioners and policymakers, future reviews can contribute to SME productivity debates by focusing on grey literature as a way of building on academic focused reviews. These were beyond the scope of this review, but do provide important insights on SME productivity.

References

- Abor, J. Y., Agbloyor, E. K., & Kuipo, R. (2014). Bank finance and export activities of small and medium enterprises. *Review of development finance*, 4(2), 97–103. https://doi.org/http://dx.doi.org/10.1016/j.rdf.2014.05.004
- Acs, Z. J., Braunerhjelm, P., Audretsch, D. B., & Carlsson, B. (2009). The knowledge spillover theory of entrepreneurship. *Small Business Economics*, 32(1), 15–30. https://doi.org/10.1007/s11187-008-9157-3
- Aldrich, H., & Auster, E. R. (1986). Even dwarfs started small: Liabilities of age and size and their strategic implications. *Research in organizational behavior*. https://doi.org/10.2139/ssrn.1497769
- Alsos, G. A., Ljunggren, E., & Hytti, U. (2013). Gender and innovation: State of the art and a research agenda. *International Journal of Gender and Entrepreneurship*, 5(3), 236–256. https://doi.org/10.1108/IJGE-06-2013-0049
- Andrews, D., Criscuolo, C., & Gal, P. N. (2016). *The best versus the rest: The global productivity slowdown, divergence across firms and the role of public policy*. OECD Publishing, Paris. https://doi.org/https://doi.org/10.1787/63629cc9-en
- Antonelli, C., Crespi, F., & Scellato, G. (2015). Productivity growth persistence: firm strategies, size and system properties. *Small Business Economics*, 45(1), 129–147. https://doi.org/10.1007/s11187-015-9644-2
- Antonioli, D., Mazzanti, M., & Pini, P. (2010). Productivity, innovation strategies and industrial relations in SMEs. Empirical evidence for a local production system in northern Italy. *International review of applied economics*, 24(4), 453–482. https://doi.org/http://dx.doi.org/10.1080/02692171.2010.483790
- Arocena, P., Núñez, I., & Villanueva, M. (2007). The effect of enhancing workers' employability on small and medium enterprises: evidence from Spain. *Small business economics*, 29(1–2), 191–201. https://doi.org/http://dx.doi.org/10.1007/s11187-006-6712-7
- Audretsch, D. B., & Belitski, M. (2020). The role of R&D and knowledge spillovers in innovation and productivity. *European Economic Review*, *123*, 103391. https://doi.org/https://doi.org/10.1016/j.euroecorev.2020.103391
- Audretsch, D. B., Kritikos, A. S., & Schiersch, A. (2020). Microfirms and innovation in the service sector. Small Business Economics, 55(4), 997–1018. https://doi.org/10.1007/s11187-020-00366-4
- Autio, E., Kenney, M., Mustar, P., Siegel, D., & Wright, M. (2014). Entrepreneurial innovation: The importance of context. *Research Policy*, 43(7), 1097–1108. https://doi.org/http://dx.doi.org/10.1016/j.respol.2014.01.015
- Bagchi-Sen, S. (2001). Product innovation and competitive advantage in an area of industrial decline: the Niagara region of Canada. *Technovation*, 21(1), 45–54. https://doi.org/https://doi.org/10.1016/S0166-4972(00)00016-X
- Bakhtiari, S. (2015). Productivity, outsourcing and exit: the case of Australian manufacturing. *Small Business Economics*. https://doi.org/10.1007/s11187-014-9604-2
- Ballestar, M. T., Díaz-Chao, Á., Sainz, J., & Torrent-Sellens, J. (2020). Knowledge, robots and productivity in SMEs: Explaining the second digital wave. *Journal of Business Research*, 108, 119–131. https://doi.org/https://doi.org/10.1016/j.jbusres.2019.11.017
- Banno, M., Piscitello, L., & Amorim varum, C. (2014). The impact of public support on SMEs' outward FDI: evidence from Italy. *Journal of Small Business Management*, 52(1), 22–38. https://doi.org/http://dx.doi.org/10.1111/jsbm.12029
- Barajas, A., Huergo, E., & Moreno, L. (2016). SME Performance and Public Support for International RJVs. *Journal of Small Business Management*, *54*(4), 1206–1228. https://doi.org/10.1111/jsbm.12221
- Barbera, F., & Moores, K. (2013). Firm ownership and productivity: a study of family and non-family SMEs. *Small business economics*, 40(4), 953–976. https://doi.org/http://dx.doi.org/10.1007/s11187-011-9405-9
- Bartel, A., Ichniowski, C., & Shaw, K. (2007). How does information technology affect productivity? Plant-level comparisons of product innovation, process improvement, and worker skills. *The quarterly journal of Economics*, 122(4), 1721–1758. https://doi.org/10.1162/qjec.2007.122.4.1721
- Baumann, J., & Kritikos, A. S. (2016). The link between R&D, innovation and productivity: Are micro firms different? *Research Policy*, 45(6), 1263–1274. https://doi.org/10.1016/j.respol.2016.03.008
- Beaver, G., & Prince, C. (2004). Management, strategy and policy in the UK small business sector: a critical review. *Journal of small business and enterprise development*. [Bradford, West Yorkshire]: https://doi.org/10.1108/14626000410519083
- Black, S. E., & Lynch, L. M. (1996). Human-capital investments and productivity. *The American Economic Review*, 86(2), 263–267. https://doi.org/10.2307/2118134
- Blancher, M. N. R., Appendino, M., Bibolov, A., Fouejieu, M. A., Li, M. J., Ndoye, A., et al. (2019). Financial

- inclusion of small and medium-sized enterprises in the Middle East and Central Asia. International Monetary Fund. https://doi.org/10.5089/9781484383124.087
- Bloom, N., Sadun, R., & Reenen, J. Van. (2016). *Management as a Technology?* https://doi.org/10.3386/w22327
- Borch, O. J., & Madsen, E. L. (2007). Dynamic capabilities facilitating innovative strategies in SMEs. *International Journal of Technoentrepreneurship*, *I*(1), 109–125. https://doi.org/0.1504/IJTE.2007.014731
- Brynjolfsson, E., & Yang, S. (1996). Information technology and productivity: a review of the literature. *Advances in computers*, 43, 179–214. https://doi.org/10.1016/S0065-2458(08)60644-0
- Bughin, J., Dimson, J., Hunt, V., Allas, T., Krishnan, M., Mischke, J., et al. (2018). Solving the United Kingdom's Productivity Puzzle. *McKinsey Global Institute Discussion Paper*, 1–38.
- Büschgens, T., Bausch, A., & Balkin, D. B. (2013). Organizational culture and innovation: A meta-analytic review. *Journal of product innovation management*, 30(4), 763–781. https://doi.org/10.1111/jpim.12021
- Business Energy and Industrial Strategy (BEIS). (2018a). Small businesses and productivity. Fifteenth Report of Session 2017-19. House of Commons.
- Business Energy and Industrial Strategy (BEIS). (2018b). Business productivity review: Government call for evidence.
- Calderini, M., & Cantamessa, M. (1997). Innovation paths in product development: An empirical research. *International Journal of Production Economics; Raising the Competitive Edge in Manufacturing*, 51(1), 1–17. https://doi.org/https://doi.org/10.1016/S0925-5273(97)00076-5
- Capello, R. (1999). SME clustering and factor productivity: A Milieu production function model. *European Planning Studies*, 7(6), 719–735. https://doi.org/10.1080/09654319908720550
- Carter, S., Mwaura, S., Ram, M., Trehan, K., & Jones, T. (2015). Barriers to ethnic minority and women's enterprise: Existing evidence, policy tensions and unsettled questions. *International Small Business Journal*, *33*(1), 49–69. https://doi.org/10.1177/0266242614556823
- Cassia, L., De Massis, A., & Kotlar, J. (2012). Exploring the effect of family control on the characteristics of SMEs in Northern Italy. *International Journal of Engineering Business Management*, 4(1), 1–8. https://doi.org/10.5772/53988
- Cicea, C., Popa, I., Marinescu, C., & Ștefan, S. C. (2019). Determinants of SMEs' performance: evidence from European countries. *Economic Research-Ekonomska Istraživanja*, 32(1), 1602–1620. https://doi.org/10.1080/1331677X.2019.1636699
- Classen, N., Carree, M., Gils, A., & Peters, B. (2014). Innovation in family and non-family SMEs: an exploratory analysis. *Small business economics*, 42(3), 595–609. https://doi.org/http://dx.doi.org/10.1007/s11187-013-9490-z
- Cocker, F., Martin, A., Scott, J., Venn, A., & Sanderson, K. (2013). Psychological distress, related work attendance, and productivity loss in small-to-medium enterprise owner/managers. *International journal of environmental research and public health*, *10*(10), 5062–5082. https://doi.org/http://dx.doi.org/10.3390/ijerph10105062
- Cohen, L. (2006). Remembrance of things past: Cultural process and practice in the analysis of career stories. *Journal of vocational behavior*, 69(2), 189–201. https://doi.org/http://dx.doi.org/10.1016/j.jvb.2006.02.002
- Colombo, M. G., Croce, A., & Grilli, L. (2013). ICT services and small businesses' productivity gains: An analysis of the adoption of broadband Internet technology. *Information Economics and Policy; ICT and Innovation*, 25(3), 171–189. https://doi.org/https://doi.org/10.1016/j.infoecopol.2012.11.001
- Criscuolo, C. (2018). Slow and divided: What can policies do to lift economies and restart engines of the growth for all? *Science, Research and Innovation Performance of the EU*. https://doi.org/doi:10.2777/14136
- Cusmano, L., Koreen, M., & Pissareva, L. (2018). 2018 OECD Ministerial Conference on SMEs. https://doi.org/https://doi.org/https://doi.org/10.1787/90c8823c-en
- Custodio, C., Ferreira, M., & Matos, P. (2017). Do general managerial skills spur innovation? *Management Science*. https://doi.org/10.1287/mnsc.2017.2828
- Dabić, M., Maley, J., Dana, L.-P., Novak, I., Pellegrini, M. M., & Caputo, A. (2019). Pathways of SME internationalization: a bibliometric and systematic review. *Small Business Economics*, 1–21. https://doi.org/10.1007/s11187-019-00181-6
- Datta, D. K., Guthrie, J. P., & Wright, P. M. (2005). Human resource management and labor productivity: does industry matter? *Academy of management Journal*, 48(1), 135–145. https://doi.org/10.2307/20159664
- De Loecker, J. (2007). Do exports generate higher productivity? Evidence from Slovenia. *Journal of International Economics*, 73(1), 69–98. https://doi.org/https://doi.org/10.1016/j.jinteco.2007.03.003

- De Loecker, J. (2013). Detecting Learning by Exporting. *American Economic Journal: Microeconomics*, 5(3), 1–21. https://doi.org/10.1257/mic.5.3.1
- Díaz-García, C., González-Moreno, A., & Sáez-Martínez, F. J. (2013). Gender diversity within R & D teams: Its impact on radicalness of innovation. *Innovation: Management, Policy and Practice*, *15*(2), 149–160. https://doi.org/10.5172/impp.2013.15.2.149
- Dora, M., Dirk, V. G., Kumar, M., Molnar, A., & Gellynck, X. (2014). Application of lean practices in small and medium-sized food enterprises. *British Food Journal*, 116(1), 125–141. https://doi.org/10.1108/BFJ-05-2012-0107
- Dora, M., Kumar, M., Goubergen, D. Van, Molnar, A., & Gellynck, X. (2013). Operational performance and critical success factors of lean manufacturing in European food processing SMEs. *Trends in Food Science & Technology*, *31*(2), 156–164. https://doi.org/http://dx.doi.org/10.1016/j.tifs.2013.03.002
- Dvouletý, O., Srhoj, S., & Pantea, S. (2020). Public SME grants and firm performance in European Union: A systematic review of empirical evidence. *Small Business Economics*. https://doi.org/10.1007/s11187-019-00306-x
- Eliasson, K., Hansson, P., & Lindvert, M. (2012). Do firms learn by exporting or learn to export? Evidence from small and medium-sized enterprises. *Small Business Economics*, *39*(2), 453–472. https://doi.org/http://dx.doi.org/10.1007/s11187-010-9314-3
- Esan, A. O., Khan, M. K., Hong, S. Q., & Naylor, C. (2013). Integrated manufacturing strategy for deployment of CADCAM methodology in a SMME. *Jnl of Manu Tech Mnagmnt*, 24(2), 257–273. https://doi.org/10.1108/17410381311292331
- Fabi, B., Lacoursire, R., Raymond, L., & St-Pierre, J. (2010). HRM capabilities as a determinant and enabler of productivity for manufacturing SMEs. *Human Systems Management*, 29(3), 115–125. https://doi.org/10.3233/HSM-2010-0720
- Fabling, R., & Grimes, A. (2014). The "Suite" Smell of Success: Personnel Practices and Firm Performance. *Industrial and labor relations review*, 67(4), 1095. https://doi.org/10.1177/0019793914546297
- Falk, M., & de Lemos, F. F. (2019). Complementarity of R&D and productivity in SME export behavior. *Journal of Business Research*, 96, 157–168. https://doi.org/https://doi.org/10.1016/j.jbusres.2018.11.018
- Falk, M., & Hagsten, E. (2015a). Export behaviour of micro firms in the swedish computer and business service industries. *Economics*, 9(32), 1–25. https://doi.org/10.5018/economics-ejournal.ja.2015-32
- Falk, M., & Hagsten, E. (2015b). Exporter productivity premium for European SMEs. *Applied Economics Letters*, 22(12), 930. https://doi.org/10.1080/13504851.2014.987914
- Fazio, G., & Piacentino, D. (2010). A Spatial Multilevel Analysis of Italian SMEs' Productivity. *Spatial Economic Analysis*, 5(3), 299. https://doi.org/10.1080/17421772.2010.493953
- Foreman-Peck, J., & Nicholls, T. (2013). SME takeovers as a contributor to regional productivity gaps. *Small Business Economics*, *41*(2), 359–378. https://doi.org/10.1007/s11187-012-9444-x
- Foreman-Peck, J., & Nicholls, T. (2015). Inter-regional mobility of entrepreneurial SMEs. *Annals of Regional Science*, *54*(1), 57–87. https://doi.org/10.1007/s00168-014-0646-9
- Gare, K., & Melin, U. (2011). SMEs need formative infrastructure for business transformation. *Journal of Enterprise Information Management*, 24(6), 520–533. https://doi.org/http://dx.doi.org/10.1108/17410391111166558
- Garengo, P., Biazzo, S., & Bititci, U. S. (2005). Performance measurement systems in SMEs: A review for a research agenda. *International journal of management reviews*, 7(1), 25–47. https://doi.org/10.1111/j.1468-2370.2005.00105.x
- Georgiadis, A., & Pitelis, C. N. (2012). Human resources and SME performance in services: empirical evidence from the UK. *International Journal of Human Resource Management*, 23(4), 808–825. https://doi.org/10.1080/09585192.2011.561236
- Gherhes, C., Williams, N., Vorley, T., & Vasconcelos, A. C. (2016). Distinguishing micro-businesses from SMEs: a systematic review of growth constraints. *Journal of Small Business and Enterprise Development*, 23(4), 939–963. https://doi.org/10.1108/JSBED-05-2016-0075
- Gibb, A. A. (2000). SME Policy, Academic Research and the Growth of Ignorance, Mythical Concepts, Myths, Assumptions, Rituals and Confusions. *International Small Business Journal*, 18(3), 13–35. https://doi.org/10.1177/0266242600183001
- Glaeser, E. L., Kallal, H. D., Scheinkman, J. A., & Shleifer, A. (1992). Growth in Cities. *Journal of Political Economy*, 100(6), 1126–1152. https://doi.org/10.1086/261856
- Godin, B. (2009). The value of science: changing conceptions of scientific productivity, 1869 to circa 1970. *Social Science Information*, 48(4), 547–586. https://doi.org/10.1177/0539018409344475
- Griffith, R., Huergo, E., Mairesse, J., & Peters, B. (2006). Innovation and productivity across four European

- countries. Oxford review of economic policy, 22(4), 483-498. https://doi.org/10.1093/oxrep/grj028
- Guedes de Carvalho, P., Nunes, P., & Serrasqueiro, Z. (2013). Growth determinants of small- and medium-sized fitness enterprises: empirical evidence from Portugal. *European Sport Management Quarterly*, *13*(4). https://doi.org/http://dx.doi.org/10.1080/16184742.2013.810296
- Gunasekaran, A., Forker, L., & Kobu, B. (2000). Improving operations performance in a small company: a case study. *Int Jrnl of Op & Prod Mnagemnt*, 20(3), 316–336. https://doi.org/10.1108/01443570010308077
- Guthrie, J. P. (2001). High-involvement work practices, turnover, and productivity: Evidence from New Zealand. *Academy of management Journal*, 44(1), 180–190. https://doi.org/10.2307/3069345
- Guzman-Cuevas, J., Caceres-Carrasco, R., & Soriano, D. R. (2009). Functional dependence and productive dependence of SMEs. *Small business economics*, *32*(3), 317–330. https://doi.org/http://dx.doi.org/10.1007/s11187-008-9115-0
- Hagsten, E., & Kotnik, P. (2017). ICT as facilitator of internationalisation in small-and medium-sized firms. Small Business Economics, 48(2), 431–446. https://doi.org/10.1007/s11187-016-9781-2
- Hall, B. H., Lotti, F., & Mairesse, J. (2009). Innovation and productivity in SMEs: empirical evidence for Italy. Small business economics, 33(1), 13–33. https://doi.org/http://dx.doi.org/10.1007/s11187-009-9184-8
- Hamilton, L. C., & Asundi, R. (2010). Technology usage and innovation: Its effect on the profitability of SMEs. *Management Research News*, *31*(11), 830–845. https://doi.org/10.1108/01409170810913033
- Harris, R., McAdam, R., & Reid, R. (2016). The Effect of Business Improvement Methods on Innovation in Small and Medium-Sized Enterprises in Peripheral Regions. *Regional Studies*, 50(12), 2040–2054. https://doi.org/10.1080/00343404.2015.1083971
- Hemert, P., Nijkamp, P., & Masurel, E. (2013). From innovation to commercialization through networks and agglomerations: analysis of sources of innovation, innovation capabilities and performance of Dutch SMEs. *Annals of Regional Science*, 50(2), 425–452. https://doi.org/http://dx.doi.org/10.1007/s00168-012-0509-1
- Henley, A., & Song, M. (2019). Innovation, internationalisation and the performance of microbusinesses. *International Small Business Journal*, 38(4), 337–364. https://doi.org/10.1177/0266242619893938
- Henriques, J., & Catarino, J. (2016). Motivating towards energy efficiency in small and medium enterprises. *Journal of Cleaner Production*, *139*, 42–50. https://doi.org/https://doi.org/10.1016/j.jclepro.2016.08.026
- Henry, C., Foss, L., & Ahl, H. (2016). Gender and entrepreneurship research: A review of methodological approaches. *International Small Business Journal*, *34*(3), 217–241. https://doi.org/10.1177/0266242614549779
- Hernandez-Espallardo, M., & Delgado-Ballester, E. (2009). Product innovation in small manufacturers, market orientation and the industry's five competitive forces: Empirical evidence from Spain. *Euro Jrnl of Inn Mnagmnt*, 12(4), 470–491. https://doi.org/10.1108/14601060910996927
- Higón, D. A., & Driffield, N. (2011). Exporting and innovation performance: Analysis of the annual small business survey in the UK. *International Small Business Journal*, 29(1), 4–24. https://doi.org/10.1177/0266242610369742
- Hogan, S. J., & Coote, L. V. (2014). Organizational culture, innovation, and performance: A test of Schein's model. *Journal of Business Research*, 67(8), 1609–1621. https://doi.org/10.1016/j.jbusres.2013.09.007
- Hottenrott, H., & Lopes-Bento, C. (2014). (International) RD collaboration and SMEs: the effectiveness of targeted public RD support schemes. *Research policy*, *43*(6), 1055–1066. https://doi.org/10.1016/j.respol.2014.01.004
- Hu, Q., Mason, R., Williams, S. J., & Found, P. (2015). Lean implementation within SMEs: a literature review. *Journal of Manufacturing Technology Management*, 26(7), 980–1012. https://doi.org/10.1108/JMTM-02-2014-0013
- Humphries, D. (2005). Fight past the jargon and find the benefits. *Ind and Commercial Training*, 37(1), 31–35. https://doi.org/10.1108/00197850510576466
- Huselid, M. A. (1995). The impact of human resource management practices on turnover, productivity, and corporate financial performance. *Academy of management journal*, *38*(3), 635–672. https://doi.org/10.5465/256741
- Ipinnaiye, O., Dineen, D., & Lenihan, H. (2017). Drivers of SME performance: a holistic and multivariate approach. *Small Business Economics*, 48(4), 883–911. https://doi.org/10.1007/s11187-016-9819-5
- Jones, J., & Corral, de Z. (2017). Doing well by doing good: A study of university-industry interactions, innovationess and firm performance in sustainability-oriented Australian SMEs. *Technological Forecasting and Social Change*, *123*, 262–270. https://doi.org/https://doi.org/10.1016/j.techfore.2016.07.036
- Jones, M., Coviello, N., & Tang, Y. K. (2011). International Entrepreneurship research (1989-2009): A domain

- ontology and thematic analysis. *Journal of Business Venturing*, 26(6), 632–659. https://doi.org/10.1016/j.jbusvent.2011.04.001
- Jones, P., Scherle, J., Pickernell, D., Packham, G., Skinner, H., & Peisl, T. (2014). Fools gold? The value of business awards to small businesses. *International Journal of Entrepreneurship and Innovation*, 15(2), 89–100. https://doi.org/10.5367/ijei.2014.0151
- Karhunen, H., & Huovari, J. (2015). R&D subsidies and productivity in SMEs. *Small Business Economics*, 45(4), 805–823. https://doi.org/http://dx.doi.org/10.1007/s11187-015-9658-9
- Kennedy, B. L. (2018). The SAGE Handbook of Qualitative Data Collection. 55 City Road, London: SAGE Publications Ltd. https://doi.org/10.4135/9781526416070
- Kersten, R., Harms, J., Liket, K., & Maas, K. (2017). Small Firms, large Impact? A systematic review of the SME Finance Literature. *World Development*, 97, 330–348. https://doi.org/https://doi.org/10.1016/j.worlddev.2017.04.012
- Kim, J. W., & Ro, Y.-J. (2017). The productivity spillover between SMEs and large firms in Korea. *Industrial and Corporate Change*, 26(6), 997–1020. https://doi.org/10.1093/icc/dtx012
- Koch, M. J., & McGrath, R. G. (1996). Improving labor productivity: Human resource management policies do matter. *Strategic Management Journal*, *17*(5), 335–354. https://doi.org/10.1002/(sici)1097-0266(199605)17:5<335::aid-smj814>3.3.co;2-i
- Kohr, C. K., Malorgio, G., & Aragrande, M. (2017). Exploring determinants of internationalisation: the case of late starters. *Intl Jnl of Wine Business Res*, 29(2), 159–177. https://doi.org/10.1108/IJWBR-05-2016-0015
- Konrad, A. M., & Mangel, R. (2000). The impact of work-life programs on firm productivity. *Strategic Management Journal*, 21(12), 1225–1237. https://doi.org/10.1002/1097-0266(200012)21:12<1225::AID-SMJ135>3.0.CO;2-3
- Kossek, E. E., & Ozeki, C. (1999). Bridging the work-family policy and productivity gap: A literature review. *Community, Work & Family*, 2(1), 7–32. https://doi.org/10.1080/13668809908414247
- Krüger, J. J. (2008). Productivity and structural change: a review of the literature. *Journal of Economic Surveys*, 22(2), 330–363. https://doi.org/10.1111/j.1467-6419.2007.00539.x
- La Rocca, M., Staglianò, R., La Rocca, T., Cariola, A., & Skatova, E. (2019). Cash holdings and SME performance in Europe: the role of firm-specific and macroeconomic moderators. *Small Business Economics*, *53*(4), 1051–1078. https://doi.org/10.1007/s11187-018-0100-y
- Laforet, S. (2013). Organizational innovation outcomes in SMEs: Effects of age, size, and sector. *Journal of World Business*, 48(4), 490–502. https://doi.org/https://doi.org/10.1016/j.jwb.2012.09.005
- Lean, J. (1998). Training and business development support for micro businesses in a peripheral area. *Journal of European Industrial Training*, 22(6), 231–236. https://doi.org/10.1108/03090599810224684
- Lejarraga, I., & Oberhofer, H. (2015). Performance of small- and medium-sized enterprises in services trade: evidence from French firms. *Small Business Economics*, 45(3), 673–702. https://doi.org/http://dx.doi.org/10.1007/s11187-015-9647-z
- Li, W., Liu, K., Belitski, M., Ghobadian, A., & O'Regan, N. (2016). e-Leadership through strategic alignment: An empirical study of small- and medium-sized enterprises in the digital age. *Journal of Information Technology*, 31(2), 185–206. https://doi.org/10.1057/jit.2016.10
- Liñán, F., & Fayolle, A. (2015). A systematic literature review on entrepreneurial intentions: citation, thematic analyses, and research agenda. *International Entrepreneurship and Management Journal*, 11(4), 907–933.
- Linna, P., Pekkola, S., Ukko, J., & Melkas, H. (2010). Defining and measuring productivity in the public sector: managerial perceptions. *International Journal of Public Sector Management*, 23(5), 479–499. https://doi.org/10.1108/09513551011058493
- Lodefalk, M., & Tang, A. (2018). The impact of hiring top workers on productivity: what is the role of absorptive capacity? *Applied Economics Letters*, 1–5. https://doi.org/10.1080/13504851.2017.1422594
- Love, J. H., & Roper, S. (2015). SME innovation, exporting and growth: A review of existing evidence. *International small business journal*, *33*(1), 28–48. https://doi.org/10.1177/0266242614550190
- Maani, K. E., Putterill, M. S., & Sluti, D. G. (1994). Empirical Analysis of Quality Improvement in Manufacturing. *Int J Qual & Reliability Mgmt*, 11(7), 19–37. https://doi.org/10.1108/02656719410738984
- Madrid-Guijarro, A., García-Pérez-de-Lema, D., & Van Auken, H. (2013). An investigation of spanish sme innovation during different economic conditions. *Journal of Small Business Management*, *51*(4), 578–601. https://doi.org/10.1111/jsbm.12004
- Mallett, O., Wapshott, R., & Vorley, T. (2018). How Do Regulations Affect SMEs? A Review of the Qualitative Evidence and a Research Agenda. *International Journal of Management Reviews*, 0(0). https://doi.org/10.1111/jimr.12191
- Mancinelli, S., & Mazzanti, M. (2009). Innovation, networking and complementarity: evidence on SME

- performances for a local economic system in North-Eastern Italy. *Annals of Regional Science*, 43(3), 567–597. https://doi.org/10.1007/s00168-008-0255-6
- Mañez, J. A., Rochina-Barrachina, M. E., Sanchis, A., & Sanchis, J. A. (2013). Do process innovations boost SMEs productivity growth? *Empirical Economics*, 44(3), 1373–1405. https://doi.org/10.1007/s00181-012-0571-7
- Martínez-Caro, E., & Cegarra-Navarro, J. G. (2010). The impact of e-business on capital productivity: An analysis of the UK telecommunications sector. *International Journal of Operations and Production Management*, 30(5), 488–507. https://doi.org/10.1108/01443571011039597
- Mayer, P., Gerber, N., McDermott, R., Volkamer, M., & Vogt, J. (2017). Productivity vs security: mitigating conflicting goals in organizations. *Info and Computer Security*, 25(2), 137–151. https://doi.org/10.1108/ICS-03-2017-0014
- McGovern, T., Small, A., & Hicks, C. (2017). Diffusion of process improvement methods in European SMEs. *International Journal of Operations and Production Management*, *37*(5), 607–629. https://doi.org/10.1108/IJOPM-11-2015-0694
- McGuinness, S., Bennett, J., & McCausland, G. (2008). Service sector SMEs and essential skill provision in the 16-25 year old labour market: evidence from Northern Ireland. *International journal of human resource management*, 19(2), 356–371. https://doi.org/10.1080/09585190701799945
- McGuirk, H., Lenihan, H., & Hart, M. (2015). Measuring the impact of innovative human capital on small firms' propensity to innovate. *Research Policy*, 44(4), 965–976.
- Middleton, K. L., & Byus, K. (2011). Information and communications technology adoption and use in small and medium businesses: The influence of Hispanic ethnicity. *Management Research Review*, *34*(1), 98–110. https://doi.org/10.1108/01409171111096496
- Middleton, K. L., & Chambers, V. (2010). Approaching digital equity: Is wifi the new leveler? *Information Technology & People*, 23(1), 4–22. https://doi.org/10.1108/09593841011022528
- Miles, Huberman, & Saldaña. (2014). *Qualitative data analysis: A methods sourcebook* (3rd ed.). Thousand Oaks, CA: Sage.
- Miles, I. (2007). Research and development (R&D) beyond manufacturing: the strange case of services R&D. *R&D Management*, *37*(3), 249–268. https://doi.org/10.1111/j.1467-9310.2007.00473.x
- Mole, K., Ghobadian, A., O'Regan, N., & Liu, J. (2004). The Use and Deployment of Soft Process Technologies within UK Manufacturing SMEs: An Empirical Assessment Using Logit Models. *Journal of Small Business Management*, 42(3), 303–324. https://doi.org/10.1111/j.1540-627x.2004.00113.x
- Motta, V. (2020). Lack of access to external finance and SME labor productivity: does project quality matter? Small Business Economics, 54(1), 119–134. https://doi.org/10.1007/s11187-018-0082-9
- Muthiah, K. M. N., & Huang, S. H. (2006). A review of literature on manufacturing systems productivity measurement and improvement. *International Journal of Industrial and Systems Engineering*, 1(4), 461–484. https://doi.org/10.1504/IJISE.2006.010387
- Naranjo Valencia, J. C., Sanz Valle, R., & Jiménez Jiménez, D. (2010). Organizational culture as determinant of product innovation. *European Journal of Innovation Management*, 13(4), 466–480. https://doi.org/10.1108/sd.2011.05627dad.006
- Nathan, M. (2014). Same difference? Minority ethnic inventors, diversity and innovation in the UK. *Journal of Economic Geography*, 15(1), 129–168. https://doi.org/10.1093/jeg/lbu006
- Niaki, M. K., & Nonino, F. (2017). Impact of additive manufacturing on business competitiveness: a multiple case study. *Jnl of Manu Tech Mnagmnt*, 28(1), 56–74. https://doi.org/10.1108/JMTM-01-2016-0001
- Niebuhr, A. (2010). Migration and innovation: Does cultural diversity matter for regional R&D activity? *Papers in Regional Science*, 89(3), 563–585. https://doi.org/10.1111/j.1435-5957.2009.00271.x
- Noke, H., & Hughes, M. (2010). Climbing the value chain. Strategies to create a new product development capability in mature SMEs. *International Journal of Operations and Production Management*, *30*(2), 132–154. https://doi.org/http://dx.doi.org/10.1108/01443571011018680
- North, D. C. (1991). Institutions. The Journal of Economic Perspectives, 5(1), 97–112.
- North, & Smallbone. (1996). Small business development in remote rural areas: The example of mature manufacturing firms in Northern England. *Journal of Rural Studies*, 12(2), 151–167. https://doi.org/10.1016/0743-0167(96)00009-5
- Nunes, Goncalves, M., & Serrasqueiro, Z. (2013). The influence of age on SMEs' growth determinants: empirical evidence. *Small business economics*, 40(2), 249–272. https://doi.org/http://dx.doi.org/10.1007/s11187-011-9363-2
- Nunes, M. B., Annansingh, F., Eaglestone, B., & Wakefield, R. (2006). Knowledge management issues in knowledge-intensive SMEs. *Journal of Documentation*, 62(1), 101–119.

- https://doi.org/10.1108/00220410610642075
- Nyikos, G., Béres, A., Laposa, T., & Závecz, G. (2020). Do financial instruments or grants have a bigger effect on SMEs' access to finance? Evidence from Hungary. *Journal of Entrepreneurship in Emerging Economies*, 12(5), 667–685. https://doi.org/10.1108/JEEE-09-2019-0139
- OECD. (2005). *OECD SME and Entrepreneurship Outlook 2005*. https://doi.org/https://doi.org/https://doi.org/10.1787/9789264009257-en
- OECD. (2017a). *Entrepreneurship at a Glance 2017*. OECD. https://doi.org/10.1787/entrepreneur_aag-2017-en OECD. (2017b). *OECD Employment Outlook 2017*. https://doi.org/https://doi.org/https://doi.org/10.1787/empl_outlook-2017-en
- OECD. (2017c). Enhancing the Contributions of SMEs in a Global and Digitalised Economy. Meeting of the OECD Council at Ministerial Level. Paris.
- OECD. (2018). *Productivity and Jobs in a Globalised World: (How) Can All Regions Benefit?* https://doi.org/https://doi.org/https://doi.org/10.1787/9789264293137-en
- OECD. (2019). *Strengthening SMEs and Entrepreneurship for Productivity and Inclusive Growth*. https://doi.org/https://doi.org/https://doi.org/10.1787/c19b6f97-en
- Oeij, P., de Vroome, E., Bolland, A., Gründemann, R., & van Teeffelen, L. (2015). Investing in Workplace Innovation Pays Off for SMEs: A Regional Innovation Initiative from the Netherlands. *International Journal of Social Quality*, 4(2), 86–106. https://doi.org/10.3167/ijsq.2014.040206
- Onkelinx, J., Manolova, T. S., & Edelman, L. F. (2016). The human factor: Investments in employee human capital, productivity, and SME internationalization. *Journal of International Management*, 22(4), 351–364. https://doi.org/10.1016/j.intman.2016.05.002
- Osiyevskyy, O., & Dewald, J. (2015). Explorative Versus Exploitative Business Model Change: The Cognitive Antecedents of Firm-Level Responses to Disruptive Innovation. *Strategic Entrepreneurship Journal*, 9(1), 58–78. https://doi.org/10.1002/sej.1192
- Ostapenko, N. (2015). National culture, institutions and economic growth: The way of influence on productivity of entrepreneurship. *Journal of Entrepreneurship and public policy*, 4(3), 331–351. https://doi.org/10.1108/JEPP-11-2014-0040
- Parisi, M. L., Schiantarelli, F., & Sembenelli, A. (2006). Productivity, innovation and R&D: Micro evidence for Italy. *European economic review.*, 50(8), 2037–2061. https://doi.org/10.1016/j.euroecorev.2005.08.002
- Paul, J., Parthasarathy, S., & Gupta, P. (2017). Exporting challenges of SMEs: A review and future research agenda. *Journal of World Business*, 52(3), 327–342. https://doi.org/https://doi.org/10.1016/j.jwb.2017.01.003
- Pittaway, L., Robertson, M., Munir, K., Denyer, D., & Neely, A. (2004). Networking and innovation: a systematic review of the evidence. *International Journal of Management Reviews*, *5-6*(3-4), 137–168. https://doi.org/10.1111/j.1460-8545.2004.00101.x
- Prowle, M., Lucas, M., Barnes, S., & Lowth, G. (2017). *Improving productivity in UK Small-medium sized enterprises: A research study*.
- Rangamohan, E., Brush, C., & Kasuganti, R. (2007). SMEs in emerging markets an overview. *International Journal of Emerging Markets*, 2(2). https://doi.org/10.1108/ijoem.2007.30102baa.002
- Rasel, F. (2016). Combining Information Technology and Decentralized Workplace Organization: SMEs versus Larger Firms. *International Journal of the Economics of Business*, 23(2), 199–241. https://doi.org/http://dx.doi.org/10.1080/13571516.2015.1106065
- Raymond, L. (2005). Operations management and advanced manufacturing technologies in SMEs: A contingency approach. *Journal of Manufacturing Technology Management*, *16*(7), 936–955. https://doi.org/10.1108/17410380510627898
- Raymond, L., & Bergeron, F. (2008). Enabling the business strategy of SMEs through e-business capabilities: A strategic alignment perspective. *Industrial Management & Data Systems*, 108(5), 577–595. https://doi.org/10.1108/02635570810876723
- Raymond, L., Bergeron, F., & Croteau, A. M. (2013). Innovation Capability and Performance Of Manufacturing Smes: The Paradoxical Effect of IT Integration. *Journal of Organizational Computing and Electronic Commerce*, 23(3), 249. https://doi.org/10.1080/10919392.2013.807714
- Raymond, L., & St-Pierre, J. (2005). Antecedents and performance outcomes of advanced manufacturing systems sophistication in SMEs. *International Journal of Operations and Production Management*, 25(6), 514–533. https://doi.org/10.1108/01443570510599692
- Ribeiro Soriano, D., & Castrogiovanni, G. J. (2012). The impact of education, experience and inner circle advisors on SME performance: insights from a study of public development centers. *Small business economics*, 38(3), 333–349. https://doi.org/http://dx.doi.org/10.1007/s11187-010-9278-3

- Romero-Jordan, D., Sanz-Labrador, I., & Sanz-Sanz, J. F. (2019). Is the corporation tax a barrier to productivity growth? *Small Business Economics*, 1–16. https://doi.org/10.1007/s11187-019-00136-x
- Ruivo, P., Oliveira, T., & Neto, M. (2015). Using resource-based view theory to assess the value of ERP commercial-packages in SMEs. *Computers in Industry*, 73, 105. https://doi.org/10.1016/j.compind.2015.06.001
- Sandulli, F. D., Baker, P. M. A., & López-Sánchez, J. I. (2013). Can small and medium enterprises benefit from skill-biased technological change? *Journal of Business Research*, 66(10), 1976–1982. https://doi.org/https://doi.org/10.1016/j.jbusres.2013.02.021
- Sandulli, F., Fernández-Menéndez, J., Rodríguez-Duarte, A., & López-Sánchez, J. I. (2012). The productivity payoff of information technology in multimarket SMEs. *Small Business Economics*, *39*(1), 99–117. https://doi.org/10.1007/s11187-010-9297-0
- Savery, L. K., & Alan Luks, J. (2004). Does training influence outcomes of organizations?: Some Australian evidence. *Journal of Mgmt Development*, 23(2), 119–123. https://doi.org/10.1108/02621710410517210
- Scuotto, V., Del Giudice, M., & Obi Omeihe, K. (2017). SMEs and Mass Collaborative Knowledge Management: Toward Understanding the Role of Social Media Networks. *Information Systems Management*, 34(3), 280–290. https://doi.org/http://dx.doi.org/10.1080/10580530.2017.1330006
- Serban, S. (2013). Impact of research and innovation on productivity of firms: empirical evidence on German SMEs. *European review of applied sociology*, *6*(7), 5–23.
- Shir, N., Nikolaev, B. N., & Wincent, J. (2018). Entrepreneurship and well-being: The role of psychological autonomy, competence, and relatedness. *Journal of Business Venturing*. https://doi.org/10.1016/j.jbusvent.2018.05.002
- Siqueira, A. C. O., & Cosh, A. D. (2008). Effects of product innovation and organisational capabilities on competitive advantage: evidence fom UK small and medium manufacturing enterprises. *International journal of innovation management*, 12(2), 113–138. https://doi.org/10.1142/S1363919608001972
- Smallbone, D. (2016). Entrepreneurship policy: issues and challenges. *Small Enterprise Research*, 23(3), 201–218. https://doi.org/10.1080/13215906.2016.1270227
- Solow, R. M. (1957). Technical Change and the Aggregate Production Function. *The Review of Economics and Statistics*, *39*(3), 312–320. https://doi.org/10.2307/1926047
- Spiegler, A. B., & Halberstadt, J. (2018). SHEstainability: how relationship networks influence the idea generation in opportunity recognition process by female social entrepreneurs. *International Journal of Entrepreneurial Venturing*, 10(2), 202–235. https://doi.org/10.1504/IJEV.2018.092716
- St-Pierre, J., & Raymond, L. (2004). Short-term effects of benchmarking on the manufacturing practices and performance of SMEs. *International Journal of Productivity and Performance Management*, *53*(8), 681–699. https://doi.org/10.1108/17410400410569107
- Struker, J., & Gille, D. (2010). RFID adoption and the role of organisational size. *Business Process Management Journal*, 16(6), 972–990. https://doi.org/10.1108/14637151011093026
- Su, Z., & Tang, J. (2016). Product innovation, cost-cutting and firm economic performance in the post-crisis context: Canadian micro evidence. *Journal of Centrum Cathedra*, 9(1), 4–26. https://doi.org/10.1108/JCC-08-2016-0009
- Thorpe, R., Holt, R., Macpherson, A., & Pittaway, L. (2005). Using knowledge within small and medium-sized firms: A systematic review of the evidence. *International Journal of Management Reviews*, 7(4), 257–281. https://doi.org/10.1111/j.1468-2370.2005.00116.x
- Tomlinson, P. R., & Fai, F. M. (2013). The nature of SME co-operation and innovation: A multi-scalar and multi-dimensional analysis. *International Journal of Production Economics*, *141*(1), 316–326. https://doi.org/http://dx.doi.org/10.1016/j.ijpe.2012.08.012
- Torre, E. D., & Solari, L. (2011). High performance work systems, technological innovations and firm performance in SME: Evidences from Italy. *International Journal of Entrepreneurial Venturing*, *3*(4), 375–391. https://doi.org/10.1504/IJEV.2011.043384
- Torrent-Sellens, J., Ficapal-Cusi, P., Boada-Grau, J., & Vigil-Colet, A. (2016). Information and communication technology, co-innovation, and perceived productivity in tourism small and medium enterprises: an exploratory analysis. *Current Issues in Tourism*, *19*(13), 1295–1308. https://doi.org/http://dx.doi.org/10.1080/13683500.2015.1029878
- Towers, N., & McLoughlin, J. (2005). Effective total quality management in the textile fashion retail supply chain: A pilot survey of the UK textile manufacturers. *Journal of the Textile Institute*, 96(2), 87–92. https://doi.org/10.1533/joti.2004.0058
- Tranfield, D., Denyer, D., & Smart, P. (2003). Towards a methodology for developing evidence-informed management knowledge by means of systematic review. *British Journal of Management*, 14(3), 207–222.

- https://doi.org/10.1111/1467-8551.00375
- Ur Rehman, N., Çela, A., Morina, F., & Sulçaj Gura, K. (2019). Barriers to growth of SMEs in Western Balkan countries. *Journal of Management Development*, 38(1), 2–24. https://doi.org/10.1108/JMD-09-2018-0273
- Uwizeyemungu, S., Poba-Nzaou, P., & St-Pierre, J. (2015). Assimilation patterns in the use of Advanced Manufacturing Technologies in SMEs: Exploring their effects on product innovation performance. *Journal of Information Systems and Technology Management : JISTEM*, 12(2), 271–288. https://doi.org/10.5748/9788599693117-12contecsi/ps-3484
- Van Beveren, I. (2012). Total factor productivity estimation: A practical review. *Journal of Economic Surveys*, 26(1), 98–128. https://doi.org/10.1111/j.1467-6419.2010.00631.x
- van Stel, A., Lokshin, B., & de Vries, N. (2019). The Effect of SME Productivity Increases on Large Firm Productivity in the EU. *Kyklos, International Review for Social Sciences*, 0(0). https://doi.org/10.1111/kykl.12198
- Welter, F. (2011). Contextualizing Entrepreneurship—Conceptual Challenges and Ways Forward. *Entrepreneurship: Theory and Practice*, 35(1), 165–184. https://doi.org/10.1111/j.1540-6520.2010.00427.x
- Welter, F., & Smallbone, D. (2011). Institutional Perspectives on Entrepreneurial Behavior in Challenging Environments. *Journal of Small Business Management*, 49(1), 107–125. https://doi.org/10.1111/j.1540-627X.2010.00317.x
- Wielicki, T., & Arendt, L. (2010). A knowledge-driven shift in perception of ICT implementation barriers: Comparative study of US and European SMEs. *Journal of Information Science*, *36*(2), 162–174. https://doi.org/10.1177/0165551509354417
- Yazdanfar, D. (2013). Profitability determinants among micro firms: evidence from Swedish data. *Int J of Managerial Finance*, 9(2), 151–160. https://doi.org/10.1108/17439131311307565